CENWP-OP-B                                                                                                                         May 8, 2008
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 18 of 2008, which covers the period from 27 April through 03 May 2008.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 183.3 to 223.4 kcfs.  Daily average project tailwater ranged from 15.8’ to 18.0’ msl.  Daily average powerhouse forebay elevation ranged from 74.0’ to 75.9’ msl.  Daily average river temperature ranged from 49°F to 51°F.  Secchi disk measurements ranged from 3.5 to 4.7.’
1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-branch, respectively, are included in the spill pattern.  Daily average spill ranged from 98.9 to 100.6 kcfs.  
1.3 Unit Operation.  Powerhouse 2 is the priority powerhouse.
1.3.1 Unit 3 has remained out of service since 14 April for a five year overhaul.

1.3.2 Unit 7 has remained out of service since 04 June 2007 for turbine rehab.

1.3.3 Unit 8 has remained out of service since 12 September 2006 for turbine rehab.
1.3.4 Until 11 was out of service from 0354 on 1559 on 30 April for field ground breaker installation.

1.3.5 Unit 12 was out of service from 0414 until 1622 on 29 April for field ground breaker installation.

1.3.6 Unit 14 was out of service from 0400 until 1150 on 28 April for field ground breaker installation.
1.3.7 Unit 15 has remained out of service since 25 November 2007due to exciter and pole failure, followed by a head gate hydraulic line rupture.
1.3.8 Unit 16 was out of service from 1157 until 1642 on 28 April for field ground breaker installation.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris.
1.5 Gatewell drawdowns are measured on Wednesdays; VBS drawdowns are measured daily on modified units.
2 ACTIVITIES

2.1 AWS Closures.  The following AWS closures occurred this reporting week:
2.1.1 FV1-1 was closed from 0955 until 1038 on 28 April for cleaning.
2.2 STS/VBS Inspections.  No STS/VBS inspections occurred this reporting week. 
Table 1.  STS and VBS inspections.  

	Unit
	Date
	Run Hours
	Comments

	11-18
	NA
	NA
	NA


2.3 Dewatering and Fish Salvage. No fish salvage activities occurred this reporting week.
3 RESEARCH  
3.1 Battelle.  BGS hydrohone installation began this reporting week.
3.2 Battelle.  Equipment installation completed for a spillway survival evaluation.
3.3 CRITFC.  Annual chinook, coho, sockeye and steelhead sampling in the AFF began 15 April.

3.4 Fish Field Unit.  Pinniped observation and scat collection continue.

3.5 NOAA- Gilbreath.  Juvenile Spring Creek fish releases in gatewell 14A continue. PIT tagged fish are being collected in SbyC and SMP samples.    
3.6 PSMFC Smolt Monitoring. Smolt monitoring activities ongoing.  Fallbacks recorded crossing the separator bars included; seven steelhead, one Chinook and one steelhead kelt (dead on arrival).  A total of 337 yearling Chinook and 25 coho were held for NOAA research projects while 84 yearling Chinook were collected for University of Washington studies this week.
3.7 USDA Wildlife Services. Sea lion hazing from the powerhouses continues.

3.8 University of Washington.  Collection of barged fish, SbyC, and run of river fish for challenge trial began on 18 April.  
3.9 WDFW Fish Counting.  Visual counts at Bradford Island and WA Shore continues.
3.10 WDFW/ODFW/CRITFC Sea lion hazing.  Hazing from boats continues.

4 FISHWAYS 

4.1 Project biologists inspected on 27 April – 03 May.
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.

	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND AND BRANCHES

	4/28, 5/1
	FG3-5
	Closed.
	Diffuser broken.
	See 4.2.4
	08-38527

	5/3
	BI diffusers
	FG3-24 open
	System in manual 
	Ops adjusted.
	

	CASCADES ISLAND

	5/3
	CI diffusers
	FG6-11 open
	System in manual
	Ops adjusted.
	

	WASHINGTON SHORE

	All week
	DSM2 
	Airburst not bursting all panels
	PLC issues.
	Electricians working on it.
	08-37503


4.2.1 FG6-10 closed.  The motor was repaired on 15 August 2007.  The shaft and gate have parted ways, however, so the diffuser will not be operational until repaired in winter 09/10.  It is OOC when TW is above 19.0’ msl.
4.2.2 PH1 electronic velocity meter is in an eddy.  Due to this placement, the velocity meter often reads less than 1.5 fps.  Readings to meet FPP criteria are taken from the south end mechanical meter.

4.2.3 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 09/10.  

4.2.4 FG3-5 closed.  The valve stem separated from the gate, repairs are scheduled for winter 09/10.  It is OOC when TW is above 16.7’ msl.

4.2.5 FV 3-7 is in manual because it is not tracking with forebay elevations correctly, and is adjusted as necessary to maintain ladder criteria.  TR 08-39938.
4.2.6 PH2 velocity meter.  The velocity meter is believed to be broken.  Repairs will not be made until winter 08/09.
4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore. Temperatures are in Fahrenheit.

4.3.1 Bradford Island temperatures are not available at this time.  The Project is waiting on a new temperature probe.  The new probe should be installed by mid-April.

4.3.2 The Washington Shore fishway temps are considerably lower than the AFF temps in the return pool.  Fisheries suspects the probe needs to be replaced.  Until such time, the AFF and WS temps will be reported.

Table 3.  Bonneville Fishway Temperatures

	Date
	WS average
	AFF 
	BI average

	4/27
	47.6
	49.8
	NA

	4/28
	46.5
	50.3
	NA

	4/29
	40.5
	49.7
	NA

	4/30
	39.2
	48.9
	NA

	5/1
	43.7
	49.8
	NA

	5/2
	46.8
	50.6
	NA

	5/3
	45.3
	51.0
	NA


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

4.5 Adult Fishways. 
4.5.1 Calibration was checked on 03 May.
4.5.2 B-Branch and Cascades Island fishway PLC are in manual.
4.5.3 Installation of the FOG sea lion exclusion bars has halted due to a broken lifting beam.  The four large orifices have bars (two at each end of the powerhouse). The stab plates remain installed in gates without exclusion bars.  
4.5.4 SLEDs and acoustic deterrents remain installed at fishway entrances.
4.5.5 The AFF is back in service.
4.6 Juvenile Fishways.

4.6.1 Orifice 15C-N was closed on 14 April to support U15 head gate removal.
4.6.2 U14 orifices were placed in manual at 1150 on 29 April until 1152 on 03 May, with 14A-S open and 14A-N closed for NOAA gatewell releases.
4.6.3 U14 orifices were closed from 0749 until 0924 on 28 Monday for STS inspection.
4.6.4 F1 and F2 orifices were moved to manual throughout the week.  When FB elevation exceeded 75.5’ msl, the orifices would be closed to prevent the channel sensor from tripping the ERG.  
4.6.4.1 F1 orifices were closed at 1055 on 01 May.

4.6.4.2 F2 orifices were closed at 1055 on 01 May.

4.6.5 Orifice 15B-N was closed at 1617 on 02 May to maintain DSM2 level.

4.7 Lamprey Fishways.  Remain winterized.
5 CONSTRUCTION
5.1 PH1 ITS Wall Removal.  Three automated gates will be installed.  Optional items include sloping the floor and removing the DSM wall.
5.2 TIE crane.  The TIE crane boom repair is on hold until funding is secured.

5.3 BGS.  Installation of the acoustic nodes continues.
6 HAZMAT, SPILLS AND CLEANUP.  Nothing to report this week.  
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 The teletype sent on 20 December, which replaced the tail water operation for chum spawning teletype on 12 December, was replaced by a teletype sent on 11 April.  This teletype, effective immediately, requests 11.5’ minimum tailwater during all hours as a soft constraint.  This request is to protect emerging chum fry.
7.2 A teletype sent on 10 April requests the Project to begin juvenile fish spill beginning at 0001 on 10 April and continuing until further notice.  Spill will be 100Kcfs (or as close as the 100K pattern allows given forebay elevation) for 24 hours a day.  The current pattern can be found in the 2008 FPP at www.nwd-wc.usace.army.mil/tmt/documents/fpp/
7.3 A teletype sent on 2 May requests special operations for the Spring Creek Hatchery fish releases. From 0600 3 May until 0559 7 May operate Powerhouse 2 as priority powerhouse.  Target 50 MW for Powerhouse 2 units up to a maximum of 57 MW as a hard constraint.  If units are required at Powerhouse 1 operate the MGR units on a first on last off basis according to FPP unit priorities. 
7.4 A teletype sent on 2 May requests U14 be operated as first priority from 0600 29 April until 0559 3 May to accommodate the NOAA gatewell tests.
JAMES R. MAHAR, P.E.

Operations Manager
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